Minutes & action items Aug. 17, 2005 TRT Barrel integration.

· Fuse box:
· Duke fuse boxes were tested all ok last week (7.5 weeks) and will be tested again later today or tomorrow. PNPI has ~2000 fuses in boxes and also saw no failure other than the known initial ones.  
· Jack will try to finish fuse boxes by the end of the month and let Chiho carry to CERN.
· Jack is going to send 1 set of fuse boxes to CERN today for use in system test. It should arrive early next week. Daniel is carrying back 1.5 set of fuse boxes from PNPI next week for the same purpose. 
· To use production fuse boxes in system test we will need adaptors to convert from SHV to the mating HV connector. The present distribution box can not be used due to opposite gender. Instead of modifying the distribution boxes, Harold suggests we make the same thing Herve is using for HV test in 154. We will need 2 of them. 
· 29 out of 30 HV connectors have been assembled in 154 and are proceeding to HV/continuity tests. The assembled connectors are still waiting for a wire to complete case grounding and interlock loop back. A total of 35 connectors are to be made but 5 are being reserved for Duke fuse boxes which already have HV cables soldered and need to be built in opposite sequence.

· Fuse box brackets are all in SR.
· After coating some of the fuse boxes at Duke, Jack found a HV cable in contact with filter capacitor has a breakdown. The HV coax is running along the HV pad. The cable ground is facing HV and separated by its outer jacket. Rick suspects the jacket may be only good for 50V or something like that. So, we should check the separation between cable and the HV pad/components and perhaps insert some Kapton as needed. On the other hand, we don’t know if this is due to interaction between the coating material and the cable jacket that weakened the insulation. Some test is needed to verify.           
· Manifold:

· The 6 o’clock quadrant of the manifold is in hand and being tested:

· The cooling manifold has been mounted on the test frame and is over being pressure tested.

· The active gas manifolds have been Pressure & Flow Tested and are in Serguei's hands for preparing the samples for cleanliness testing. 

· John expects to start mounting the manifolds Thursday or Friday. 

· Cooling tubes in 6 o’clock quadrant are all installed.

· B&E has all the parts they need and is doing the final assembly of the mounting ears. They expect to ship the last 2 quadrants either today or tomorrow. 
· Other items:
· The installation of the 3rd quadrant electronics is expected to finish next week and the last quadrant ~a week after. 
· Stack 7 is expected to join stack 8 in cosmic ray data taking next week. 

· The Sep. 15 moving date now seems pretty tight and was questioned. We do need to finish sector test and a couple of other things. There are also structural works that need to be done before the barrel can move.
